Author's response to reviews

Title: Protein Secretion in Human Mammary Epithelial Cells following HER1 Receptor Activation: Influence of HER2 and HER3 Expression

Authors:
Yi Zhang (yzhang81@uthsc.edu)
Rachel M Gonzalez (rmgher@uw.edu)
Richard C Zangar (richard.zangar@pnl.gov)

Version: 6 Date: 2 February 2011

Author's response to reviews:

February 2, 2011

RE: BMC Cancer MS: 1117246558431675

Dear Editors of BMC Cancer,

We would like to resubmit an original research article entitled “Protein Secretion in Human Mammary Epithelial Cells following HER1 Receptor Activation: Influence of HER2 and HER3 Expression” for publication. As requested, we have made minor editorial changes to the manuscript. None of these changes result in substantive changes in the data or resulting conclusions of the study.

Best Regards,

Richard C. Zangar, Ph.D.
790 Sixth Street
Pacific Northwest National Laboratory
Richland, WA 99352
T: 1-509-371-7301
F: 1-509-371-7304
richard.zangar@pnl.gov